# Review



# On the Simulation of LoRaWAN Networks: A Focus on Reproducible Parameter Configuration

Marianna Gois de Campos<sup>(D)</sup>, Lahis Gomes de Almeida, Luiz Eduardo Mendes Matheus, and Juliana Freitag Borin<sup>\*</sup><sup>(D)</sup>

Institute of Computing, University of Campinas, Campinas, Brazil E-mail: juliana@ic.unicamp.br

Received: 29 February 2024; Revised: 8 May 2024; Accepted: 20 May 2024

**Abstract:** In the past decade, with the rapid proliferation of Internet of Things (IoT) devices new applications have emerged in diverse fields such as agriculture, smart cities, and healthcare. In this context, LoRaWAN, a Low-Power Wide Area Network protocol, has gained significant adoption. Its advantages, including low-power consumption and long-range communication capabilities, make it an ideal solution for IoT communication. Simulation tools have played a crucial role in facilitating experimentation and advancing the implementation of the LoRaWAN protocol. However, effectively modelling these experiments present a significant challenge. Through a comprehensive survey of the literature on LoRaWAN, we have identified variations in parameter configuration among different works and observed a lack of sufficient information in most of them, thereby impeding result reproducibility. This paper offers a comprehensive overview of LoRaWAN technology, encompassing its architecture and underlying technologies such as LoRa and LPWAN. Furthermore, we present and compare the main simulation tools currently available for LoRaWAN, discuss the parameters that significantly impact the performance of a LoRaWAN network, and explore how researchers have configured these parameters to model their simulation experiments.

Keywords: LoRa, LoRaWAN, simulation tools, simulation parameters, reproducibility

# 1. Introduction

During the last decade, the Internet has gone through a revolution in terms of network organization and paradigms. Nowadays, we are able to design and visualize everyday devices as connected things. The paradigm of the Internet of Things (IoT) has leveraged new applications and the need to rework the traditional network layers. This promising scenario has attracted the attention of both academy and industry [1].

This new era of millions of internet-connected battery-powered devices presents several challenges, including communication range, scalability, network flexibility, and power management. These challenges have given rise to innovative solutions at both the physical and link layers [2]. One promising solution for meeting these challenges is Low-Power Wide Area Network (LPWAN), which aligns well with the requirements of IoT applications. LPWAN protocols provide low data throughput, ensuring low power consumption [3]. Additionally, LPWAN technologies are capable of transmitting data between devices at distances of up to 10 km [4].

There are a wide range of applications based on LPWAN, varying from healthcare and agriculture [5–7] to disaster planning using geographical and meteorological sensing [8]. Different communication protocols were developed to meet the requirements of LPWAN, such as Sigfox [9], LoRaWAN [10] and LTE-M [11]. The next few years will see even greater developments in wireless networks, with the roll-out of 5G networks [12] and the emergence of 6G [13]. This major revolution in wireless communication will bring new technologies at all

Copyright ©2024 Marianna Gois de Campos, et al. DOI: https://doi.org/10.37256/cnc.2120244496 This is an open-access article distributed under a CC BY license (Creative Commons Attribution 4.0 International License) https://creativecommons.org/licenses/by/4.0/ levels, from the physical layer to real-time applications. As a result, communication technologies, including LoRaWAN, will be impacted with new paradigms such as network slicing, cloud Radio Access Network (cloud-RAN), and heterogeneous networks becoming part of the future landscape [14].

The use of simulation tools has proven to be a low-cost solution to collect data in different scenarios, providing studies of potential projects and being essential for the development of wireless networks. In addition, during the planning of a project, simulation tools allow us to ensure that the correct technology has been chosen [15]. This paper aims to provide a comprehensive overview of LoRaWAN architecture, the main simulation tools for this technology, and a comparison of how crucial parameters are configured in both simulation and real-world experiments found in the literature. By addressing the lack of detailed methodology explanations in much of the experimental work in LoRaWAN, this paper will assist researchers in selecting appropriate simulation tools and configuring simulation parameters effectively. We are confident that the insights presented in this paper will greatly benefit researchers in this field.

## 1.1 Related Work

Although LoRaWAN is a relatively new technology, it has gained significant popularity along with its surrounding technologies (LoRA, LPWAN) in recent years. This is due to factors such as ease of deployment, scalability, accessibility, and rapid industrial adoption [16]. The academic community has responded by producing numerous works in the area. To gain a comprehensive understanding of how this technology is explored in the literature and to compare it to related topics, recent surveys were selected and analyzed. Particular attention was paid to the simulation aspects of LoRaWAN technology. To provide a comprehensive overview of recent surveys in the field, we have classified them into three categories: (i) those covering various LPWAN technologies, (ii) those specifically exploring the LoRa PHY layer, and (iii) those concentrating on LoRaWAN.

LPWAN: Studies on LPWAN aim to characterize the different technologies that use this type of network, such as Sigfox, LoRaWAN and NB-IoT. For example, Ayoub et al. [17] explore the various aspects of LPWAN network technologies and their suitability for applications with mobility requirements. Another study by Jonathan et al. [18] compares LoRaWAN with other proprietary LPWAN network technologies. Aldahdouh et al. [19] analyses LPWAN as a 5G candidate, in comparison with other emerging technologies (NB-IoT, Sigfox, and LTE-M). Triantafyllou et al [20] provides a more general overview by comparing LPWAN (including LoRaWAN) with other wireless communication technologies.

**LoRa**: Studies on LoRaWAN often focus on its proprietary physical layer mechanism, LoRa. Lavric et al. [21] provide an overview of both LoRa and LoRaWAN technologies, with a focus on LoRa modulation and the main challenges of the technology. Saari et al. [22] conducted a survey to categorize the current state-of-the-art of LoRa-based technologies. Recently, Sarker et al. [23] took a new approach to LoRa by analyzing the literature and proposing an architecture to integrate edge computing features into IoT-based devices.

**LoRaWAN**: The literature on LoRaWAN technology has seen a number of significant contributions. Haxhibeqiri et al. [24] conducted a comprehensive analysis of the technology, exploring its network layer and proposing possible protocol enhancements. Khutsoane et al. [25] reviewed key parameters affecting the feasibility of LoRaWAN in wireless fields. Adelantado et al. [26] analyzed the limitations of the technology. Several recent studies have focused on exploring use cases and application scenarios for LoRaWAN [26–28]. Additionally, the technology has been evaluated in terms of energy efficiency [29] and feasibility in a massive IoT world [30]. LoRaWAN is also studied as a strong candidate and enabler for several applications, such as Tracking [17,22,25–28], Health Care [23,24], Agriculture [30], Smart Vehicles [31,32] and Smart Cities [33,34]. In the area of LoRaWAN simulations, Almuhaya et al. [35] and Bouras et al. [36] provided an overview of the technology, analyzing current simulation tools based on aspects such as operating system, language, community support, installation requirements, and number of previous work that used such tools. Another study [37] focused in one specific simulator (ns-3) and explored the different LoRaWAN simulation modules available for it, considering features and limitations.

Unlike prior work [35–37], this paper offers a thorough analysis of LoRaWAN, simulation tools, and propagation models and focuses on examining the ways in which simulations are conducted by researchers based on important parameters.

# **1.2** Contributions

This survey offers a novel perspective on the literature concerning LoRaWAN technology and simulation experiments. Its main contributions include:

- providing an overview of LoRaWAN, with a focus on its architectural components, and introducing a classification of Network Servers into four distinct categories;
- reviewing and comparing the primary simulation tools available for LoRa and LoRaWAN;
- reviewing the main channel propagation models used for LoRa simulation;
- reviewing the key parameters that have a significant impact on the performance of LoRaWAN networks and comparing how authors of both testbed and simulation-based studies configure their values.

## **1.3** Paper Organization

Figure 1 presents the structure of this paper. The remaining sections are organized as follows: Section 2 provides an overview of LPWAN networks; Section 3 presents a comprehensive overview of the LoRaWAN network architecture; Section 4 review and compares the main simulation tools for LoRa and LoRAWAN; Section 5 presents the channel propagation models commonly used for LoRaWAN simulation; Section 6 explores the key parameters that significantly impact the performance of LoRaWAN networks and their configuration in experimental scenarios; finally, Section 7, draws conclusions and outlines future research directions.



Figure 1. Structure of this survey.

# 2. LPWAN Overview

Low-Power Wide Area Network (LPWAN) is a type of WAN that enables long-range communication between devices while conserving power by transmitting at a low bit rate [38]. Unlike a wireless WAN, which is designed to connect users across a geographic region, LPWAN focuses on connecting IoT devices. Figure 2 provides a brief comparison between LPWAN and other communication technologies.



Figure 2. LPWAN and other network communication technologies [39]

LPWAN has garnered significant attention in recent years due to its rapid adoption in IoT deployment for businesses around the world. This trend is expected to continue, as IoT market reports predict that the number of connected IoT devices will increase from 7 billion in 2018 to over 22 billion in 2025, with LPWAN devices growing at a rate of more than 100% annually [40].

The communication range in LPWAN is typically 5 to 40 km, depending on environmental conditions, which results in a reduced number of access points when compared to other technologies. The low power consumption of LPWAN also enables battery-powered devices to operate for more than 10 years. Additionally, end nodes are usually much less expensive due to their simple design and low-cost antennas.

LPWAN technology has become an important enabler in the growth and evolution of the IoT market, offering solutions for long-range, low-power communication between devices. The different LPWAN solutions available in the market, such as SigFox, LoRAWAN and NB-IoT, have their own technical features and tradeoffs, making them suitable for specific IoT applications. Further analysis and comparisons, such as the one conducted by Mekki et al. [41], help in understanding the distinctions among these solutions. In this paper, our focus is on LoRaWAN technology.

# 3. LoRAWAN Overview

LoRaWAN is a communication protocol built on top of the LoRa Physical Layer and is a leading LPWAN technology in the IoT industry. One of the reasons for its broad adoption is that it uses an unlicensed range of radio frequency spectrum [22], which avoids the spectrum congestion of other well-known frequency ranges, such as the one used by cellular network companies. LoRaWAN addresses the challenge of region restrictions by being specified for a number of bands for each region, which directly impacts the backend infrastructure.

The modulation and physical layer protocol used result in a long-range communication with low data rate, enabling low energy consumption for battery-operated IoT devices [18]. The LoRa Alliance [42], a non-profit association of companies, standardizes the MAC features of LoRa, promoting the development and adoption of this technology by means of an open standard for LPWAN solutions.

Table 1 presents a comparison of LoRAWAN and other LPWAN technologies. As we can see, battery lifetime, power efficiency, and interference immunity are some of the aspects in which LoRaWAN stands out. This, in addition to the long range coverage and the community involvement, are the main reason behind the huge adoption of this technology for IoT solutions [42]. The following sections provide more information on the technical aspects of LoRaWAN.

Characteristics	LoRaWAN	NB-IoT	LTE-M	SigFox
Modulation	CSS	BPSK, QPSK	BPSK, QPSK, 16 QAM, 64 QAM	D-BPSK
Spectrum	Unlicensed	Licensed LTE	Licensed LTE	Unlicensed
Bidirectional Communication	Yes	Limited	Yes	Limited
Data Rate	< 50 Kbps	< 200 Kbps	< 1000 Kbps	< 100 bps
Coverage	Urban: 2-5 km	Urban: 1 km	Urban: 1 km	Urban: 10 km
	Rural: 45 km	Rural: 10 km	Rural: 10 km	Rural: 50 km
Bandwidth	125, 250, 500 KHz	200 kHz	1.400 MHz	100, 600 Hz
MAC	ALOHA	SC-FDMA	SC-FDMA	TDMA
Transmission Power	14-20 dBm	20-23 dBm	23 dBm	14 dBm
Encription	Yes	Yes	Yes	Yes

Table 1. Comparison of popular LPWAN technologies

# 3.1 LoRA

LoRaWAN is built on top of the proprietary physical layer technology known as LoRa (Long-Range) [42]. This technology utilizes Chirp Spread Spectrum (CSS) modulation for data encoding. In the field of communication technology, there are several ways to encode data at the physical level using different modulation mechanisms. These mechanisms differ in the way the carrier signal is manipulated to represent digital data. Some of the most well-known modulation techniques include:

• Amplitude Shift Keying (ASK): the amplitude of the carrier signal is altered to represent the binary data of 0s and 1s;

- Frequency Shift Keying (FSK): the frequency of the carrier signal represents the digital data being transmitted;
- Phase Shift Keying (PSK): the phase of the carrier signal is modified to represent either 0 or 1;
- Chirp Spread Spectrum (CSS): is a specific spread spectrum mechanism where a signal is spread in the frequency domain. A chirp is a signal that increases (up-chirp) or decreases (down-chirp) its frequency over time, and the original message is encoded in these chirps.

As previously noted, LoRa is based on CSS and is a proprietary modulation scheme patented by SemTech [43]. The unique feature of this modulation technique is that it provides long-range communication with a low energy consumption, making it ideal for battery-powered devices used in IoT networks [21]. In urban environments, the range of LoRa signals can vary from 10 to 15 Kms. However, the maximum distance that a LoRaWAN packet has traveled was 766 Kms during an experiment where a packet sent from Ariza, Spain was received by a gateway in Grenoble, France.

Figure 3 illustrates an example of a LoRa signal using different Spreading Factors (SF). As can be seen, the signal continuously spans the channel and changes its frequency over time. The SF determines the length of the chirp [44], with an SF of 7 representing the shortest transmission time and an SF of 12 representing the longest transmission time for the signal. Generally, a lower spreading factor results in a shorter transmission time but improved resistance to interference, while a higher spreading factor results in a longer transmission time but reduced resistance to interference.



Figure 3. LoRa Chirp Spread Spectrum spectrogram for different spreading factors [44]

Despite being a proprietary technology, there have been efforts in the research community to decode the LoRa signal and gain a deeper understanding of the modulation scheme. For instance, Knight et al. [45] have explored the modulation and encoding components of LoRa and have presented an open-source version of LoRa based on their findings.

#### **3.2** Communication Architecture

This section describes the network architecture of LoRaWAN. LoRaWAN networks employ a star-of-stars topology. In essence, end devices transmit data to gateways, which then forward it to the network server. The collected data can then be accessed by an application server, making it available to the end application. Conversely, the network server can also send data to end nodes, which will be forwarded to them through the gateways [42]. An overview of the architecture is presented in Figure 4.

In order to comprehend the intricacy of each component in the LoRaWAN architecture, a bottom-up approach will be utilized in the following subsections, beginning with the end node devices and progressing towards the application server.



Figure 4. LoRaWAN architecture overview.

## **3.2.1 End Devices**

End devices, also referred to as nodes, are responsible for collecting and transmitting the useful data to the gateways via LoRa. These nodes are not tied to a specific gateway, so when data is transmitted, it is received by every gateway within the communication range of the end device, and then forwarded to the network server [42].

There are different implementations of LoRaWAN devices, with varying differences in how the link channel is accessed, which has a direct impact on energy consumption. The LoRa Alliance defines three classes of devices: Class A, Class B, and Class C.

Class A is a mandatory class for every end device. It supports bidirectional communication, allowing for both sending messages to the gateway and receiving messages from it. Class A devices send messages at any time. After the transmission, the device opens two receive windows at 1s and 2s to listen for server replies.

Class B extends the capabilities of Class A devices by adding beacon synchronization with the network server, enabling the scheduling of additional receive windows to receive downlink packets.

Class C devices have an always-open receive window, unless they are transmitting a message. This makes communication from the network server to the end devices more efficient, but also results in increased energy consumption due to the always-open receive window.

#### 3.2.2 Gateway

Gateways serve as intermediaries in the transfer of data from end devices to the Network Server. This process involves the collection of messages sent by LoRaWAN-compliant devices and their subsequent transfer to high-bandwidth networks such as 4G, Wi-Fi, or Ethernet.

An analogy can be drawn between gateways and routers in a Wi-Fi network. However, gateways are equipped with a LoRaWAN concentrator, which can take the form of a board that can be connected to embedded devices such as Raspberry Pi [46] or BeagleBone Black [47]. There are also LoRaWAN gateways available on the market, such as those produced by MikroTik (https://mikrotik.com/product/wap\_lr8\_kit##fndtn -specifications) and Cisco (https://www.cisco.com/c/en/us/products/routers/wireless-gateway-lorawan/index.ht ml).

The level of control required determines the complexity of the gateway, and thus they can be classified into two categories:

Minimal Firmware: are usually low cost and easy to set up; they are simple devices which are only capable of forwarding the packets;

**Operating System**: are typically more sophisticated devices that run a complete operating system, with the packet forwarder being one of the tasks it performs. This allows the network administrator to choose the software used for packet forwarding, making packet management simpler and more flexible.

#### 3.2.3 Network Server

LoRaWAN devices operate differently from the ones based on the IP protocol stack, utilizing an alternate protocol to transmit data to gateways. As a result, a separate component must handle the processing and IP protocol-based routing of that data.

#### Volume 2 Issue 1|2024| 153

Network Servers form the backbone of a LoRaWAN network, taking on the role of integrating all network com-ponents, from devices to applications [42]. Network Servers vary in implementation and cater to various industry requirements. After conducting a review, we have classified Network Servers into four categories: Open-Source Network Server, Gateway Embedded Network Solution, Community Network Server, and Private Network Server.

**Open-Source Network Server**: provides full control over its source code and often relies on contributions from its user community. The benefit of this type of solution is the ability to customize, certify, and manage the entire network to meet specific application requirements. However, it comes with a high level of complexity, including the need for technical expertise for installation and configuration. Additionally, there may be a lack of specialized support, and some LoRaWAN features may not be fully developed. ChirpStack [48] is an example of an Open-Source Network Server.

**Gateway Embedded Network Solution**: is a type of Network Server that is integrated into the gateway itself. With this solution, the user typically purchases a gateway and can easily set up and deploy the Network Server, allowing for the management of devices connected to the gateway and data transmission to the application server. UrsaLink [49] is an example of this type of Network Server. However, in complex real-world deployments, this solution may have limitations such as scalability issues, limited LoRaWAN features (such as geoservers), the need for physical access to the gateways, a lack of monitoring mechanisms, and limited control over the network [50].

**Community Network Server**: is a decentralized solution where Network Servers are located globally and are available for free use by any user. This allows users to manage their devices and gateways using the infrastructure provided by the Network Server. This type of solution is usually offered by IoT companies, but may come with limitations such as limitations on the number of gateways/devices, on the access to logging and monitoring features, quality of service (QoS), and others. However, it provides users with a fast and straightforward way to access the LoRaWAN network, making it a suitable option for academic research and proof-of-concept projects. Loriot [51] is a well-known example of a Network Server solution that provides a Community Server option for its members.

**Private Network Server**: provides increased control and security, tailored to the specific requirements of the client. Its primary purpose is to offer full control over the network, customized to meet the specific needs of the client, as determined by their subscription.

#### **3.2.4 Application Server**

The data generated by end devices is transmitted through the network to reach the application server, which serves as the final destination. This server provides a platform for managing devices, applications, and organizations, and can also be accessed by other end applications using various methods. Typically, an application server is integrated into a network server solution and offers multiple mechanisms for clients to retrieve the data from sensors. Below, we give a brief overview of two popular connectivity mechanisms that are used, Message Queuing Telemetry Transport (MQTT) and HyperText Transfer Protocol (HTTP) [52].

MQTT protocol is an ISO standard for communication based on a publish-subscribe model [53]. It works as an alternative to traditional client-server communication, in a sense that in MQTT, publishers and subscribers never communicate directly to each other. The communication is interfaced by the broker, a third element introduced by MQTT that is responsible for managing and filtering the messages going through it. This decoupling logic gives the protocol flexibility in terms of time, space and synchronization. MQTT is optimised for remote communications on low bit rate networks, making it one of the most popular communication protocol for IoT solutions [54]. In MQTT, both publishers and subscribers are clients, while the broker assumes the server rule. A client is responsible for a series of operations, such as opening/closing network connection with the server, subscribing/unsubscribing to receive application messages from the server. The messages transmitted contain payload data, QoS, collection of properties and a topic name. The server (broker) is responsible for managing the communication between the MQTT clients (subscribers and publishers), which includes receiving, filtering, and sending the data. Authentication and authorization is also part of the broker's responsibility. In addition, the MQTT broker can be embedded into the gateway or in the cloud, increasing the flexibility of communication. From LoRAWAN perspective, MQTT is a fine selection to tie the communication between all the clients, which could be represented by LoRAWAN Gateways and applications. The sensors can forward their data to the gateways using LoRA, and those gateways will translate the messages into MQTT messages and forward them to the broker, hosted on the application server, which will manage it and send to the corresponding clients.

HTTP is a widely-used protocol that operates on the client-server model. It is utilized to obtain data from the Application Server, which can be connected to various services. If the Application Server provides this capability, the user can retrieve specific data through GET requests triggered by specific events, for instance.

Additionally, the integration between the end application and the application server is achieved through Application Programming Interfaces (APIs). The network server provider, such as ChirpStack, may offer APIs, such as Representational State transfer (RESTful) and Google Remote Procedure Call (gRPC), as part of their open-source application server [48]. Alternatively, the user has the option to develop their own application and retrieve information from the Application Server as needed.

The application server can also include security features. For instance, the application server can encrypt the data being sent to it. Additionally, each device has a unique application key, which can be registered with the application server [55].

# 4. LoRA and LoRaWAN Simulation Tools

This session is devoted to introducing the primary simulation tools for LoRa and LoRaWAN that are available in the literature. The objective is to provide a comprehensive overview of these tools and to gain insights into potential areas for improvement.

## 4.1 *NS-3*

NS-3 is a popular simulation tool, specially valuable to educational and academic purposes. It is a discreteevent network simulator that is implemented in both C++ and Python and it is backed by a robust community support and documentation. Furthermore, it encompasses a wide range of protocols, both Internet-based and otherwise [56]. NS-3 is created as a collection of libraries or "models" that simulate various aspects of a network. These include channel propagation, mobility of end nodes and gateways, PHY and MAC layers, and more. The simulation results can be viewed through animations and external tools, as NS-3 has a command-line based user interface [56].

The modular architecture of NS-3 is another key advantage that allows users to enhance the software by fixing bugs or adding new features. Scientific contribution plays a crucial role in technological advancement and has great significance in the networking industry. The more collaboration among researchers, the quicker new tools and updates emerge. NS-3 serves as a prime example, with its frequent release of updated models and protocols each year, reflecting the active involvement of its user community [37].

In this section, we will examine some of the LoraWAN networking modules designed for the NS-3 simulator as described in [37]. We will cover the main aspects of each module. It is noteworthy that all four modules only simulate class A devices.

#### 4.1.1 Module I

The purpose of this module (https://github.com/networkedsystems/lora-ns3) is to promote studies and advancements in LoRa technology with class A devices through NS-3 simulation. It implements features such as support for distributed gateways connected via an IP network to a controlling server, classes for easy application installation and new MAC layer commands, and a class to monitor the energy consumption of LoRa protocol's physical layer states (LoraRadioEnergyModel) [57].

The module's architecture, implemented classes, and public code repository are thoroughly explained in its publication, along with demonstrations of three simulation scenarios with varying parameters. The authors emphasize the use of the module to address issues such as the impact of different spreading factors, packet interference, network reliability and scalability, among others.

#### 4.1.2 Module II

Van den Abeele et al. [58] propose combining a LoRa error model with the LoRaWAN protocol's MAC layer to create a module for NS-3. The primary objective of their study is to examine the scalability and capacity of LoraWAN networks as the number of gateways and devices grows. The error model serves as an interference model, and its formulation involves intricate simulations of bandwidth and bit error rates.

The proposed module (https://github.com/imec-idlab/ns-3-dev-git/tree/lorawan) facilitates the simulation of scenarios involving multiple gateways and devices, different types and patterns of traffic, variations in data rates, retransmissions, and reception parameters. The paper [58] shows that increasing the density of gateways can reduce the negative impact of downstream traffic on the delivery rate of confirmed upstream traffic, but it

cannot completely eliminate it. The authors also provide a comprehensive description of the module, access to its repository, and some example code, although some of the code is not functional.

#### 4.1.3 Module III

The module (https://github.com/drakkar-lig/lora-ns3-module) proposed by To and Duda [59] was created as part of their research aimed at enhancing the performance of LoRa devices while maintaining their low power consumption and improving the accuracy of LoRaWAN network simulations in NS-3. To validate the module, the authors compared the results obtained from the module with metrics from a real-world testing environment.

The enhancements made to the module to minimize power loss include the implementation of the Carrier Sense Multiple Access (CSMA) and CSMA-x protocols. CSMA involves checking if the channel is being used by another transmission before attempting to send a packet, a technique commonly referred to as Listen Before Talk (LBT). The CSMA-x protocol extends the CSMA approach by requiring the device to test the channel for a specified interval, x, and if there is no activity, it can attempt a transmission.

The authors claim that the module accurately simulates the capture effect, which reduces the rate of packet loss due to collisions. The simulation results obtained with the implementation of CSMA show a reduction in the collision rate, although there is a moderate increase in the power consumption of the devices [59]. However, the article lacks actual description of the CSMA and CSMA-x protocols implementation, and the module code only allows for a single gateway, with limited documentation.

#### 4.1.4 Module IV

The module (https://github.com/signetlabdei/lorawan) by Magrin et al. [60] was developed with the aim of evaluating the performance of a LoRaWAN network in an urban environment. It is based on the consolidation of two key models: the LoRa physical layer, which represents the behavior and transmission of chirps, and the MAC layer.

This module includes the following features: simulations with multiple gateways and devices that have infrequent access to a wireless channel, the Adaptive Data Rate (ADR) algorithm, a basic network server, and functions to calculate statistics after the simulation. It is important to note that the module does not support network configurations outside of Europe, and there are no models to evaluate interference between partially overlapping channels or to implement certain MAC layer commands.

In their initial study [60], the authors highlight a significant improvement in coverage and uplink reliability when simulating the LoRaWAN network with a larger number of gateways. The results also showed that LoRaWAN has higher throughput compared to ALOHA due to the partial orthogonality of its spreading factors.

The authors have also published additional related work [61,62], which present solutions to additional problems that have been integrated into the source code. This module is constantly being improved, with a large user base and extensive documentation, including code examples and a NS-3 user mailing list. As a result, it is currently the most advanced module of its kind.

## 4.2 Flora

Flora (https://flora.aalto.fi/) was created as part of a research project aimed at exploring adaptive mechanisms in LoRA parameter configuration. It is a simulation framework written in C++ and based on the popular network simulation tool OMNeT++. Flora offers a range of features, including the design of LoRa networks (end nodes, gateways, and network servers), bidirectional communication, PHY and MAC layers, as well as the ability to simulate a backhaul network and evaluate its energy consumption [63].

The user interface of Flora is inherited from OMNeT++ and allows the network to be visualized. Additionally, example scenarios are included as part of the simulation tool.

#### 4.3 LoraSim

LoraSim (https://github.com/adwaitnd/lorasim) is among the earliest simulations tools for LoRa technology [64]. It uses SimPy, a discrete event simulation framework built on the Python programming language. In this simulator, end nodes can be either randomly positioned or arranged in a two-dimensional grid. The tool offers four distinct simulation types, each featuring different network properties, such as node and antenna configurations. The number of end nodes, LoRa networks, gateways, and simulation duration are among the parameters that can be altered. Additionally, the simulation produces several key outputs, including the Data Extraction Rate (DER) and Network Energy Consumption (NEC).

# 4.4 Phy Simulator

The Spreading Factor determines the number of chirps that are transmitted per second from the device to the gateway. The network takes into account the environmental conditions between the device and gateway to set the spreading factor for communication, which ranges from 7 to 12. A lower SF means more chirps per second, while a higher SF indicates fewer chirps per second [65].

PhySimulator evaluates the ability to receive two overlapping LoRa transmissions modulated with different SFs. The purpose of the PhySimulator authors is to study the SFs used by LoRa, which are nearly orthogonal, leading to collisions and incorrect reception in both close and distant conditions. The simulator is implemented in MATLAB and lacks a graphical interface. However, the user has the ability to modify parameters such as the number of load bits and bandwidth [36].

#### **4.5** Comparison of Simulation Tools

In this section, we offer a comparative assessment of the leading simulation tools currently available. Previous works such as [35] and [36] focus on comparing features like simulation tool license type, programming language, and documentation. This work focuses on comparing the primary simulation tools along with their configurable parameters, as can be seen in Table 2. We extend [37] contributions by showing and discussing configurable parameters advantages and disadvantages. This information can be useful for researchers to select or contribute to a particular tool.

All the tools listed in Table 2 are open source. The parameter values in the table were obtained through code examples from each simulator and their values can be adjusted to suit the specific scenario being studied. However, if any changes are made to certain parameters such as device class or regional frequency-related parameters (such as frequency, bandwidth, transmission power, or packet size), it is necessary to ensure that the code meets the technical requirements of these parameters. On the other hand, modifications to other parameters such as the number of end devices (ED no.) and gateways (GW no.), coordinates of end devices and gateways, propagation models, and packet intervals (e.g., in order of seconds, minutes, hours) can be made without introducing new functions or classes to the code.

Simulator	ED no.	GW no.	ADR	Check Collision	Propagation Models	Frequency	Packet size	Time Between Packets	Energy Model	Last Update
NS-3 Module I	2,400	1	√	$\checkmark$	Log Distance + Building Propagation + Correlated Shadowing	868MHz 915MHz	51 B	600 ms	Radio Energy consumption	Jan 31, 2022
NS-3 Module II	500	1	$\checkmark$	$\checkmark$	Okumura-Hata + Nakagami	868MHz	51 B	120 min	Radio Energy consumption	Mar 31, 2020
NS-3 Module III	30	1	$\checkmark$	$\checkmark$	Log Distance	868MHz	21 B	600 us	×	Mar 19, 2017
NS-3 Module IV	1	1	Х	Х	Log Distance	868MHz	13 B	100 s	Х	Dec 9, 2021
Flora	100, 1000	1,2	$\checkmark$	$\checkmark$	Log Normal Shadowing, Okumura-Hata, Oulu	868MHz	-	1 s	Radio Energy consumption	Jun 9, 2022
LoraSim	200, 1.6k	1-8, 24	$\checkmark$	$\checkmark$	Log Normal Shadowing	868MHz	20 B	1E6 ms	Radio Energy consumption	Dec 19, 2019
Phy Simulator	1	1	-	$\checkmark$	-	868MHz	20 B	7200 s	Х	-

Table 2. Simulation tools with default main parameters.

The presented table illustrates that not all of the simulation tools support the Network Server feature. While modules II, III, and IV consider it an enhancement in the framework, others such as LoraSim and Phy Simulator do not aim to incorporate this feature as they focus on the physical layer. Regarding collision checking and ADR usage, nearly all tools provide the option to enable or disable these features. With respect to channel modeling, the majority of frameworks use the Log Distance channel propagation model. In contrast, Module I applies shadowing models such as Correlated Shadowing and Building Propagation Loss. The other modules (II, IV, Flora, LoraSim) employ the Log Distance model with its shadowing constant (Log Normal Shadowing).

The simulation tools mostly use the European frequency (868MHz), with the exception of Module I, which also offers the option to simulate with the North American frequency (915MHz). In terms of energy modeling, only a few tools (Module I, Module II, Flora, and LoraSim) take into account the power consumption of the LoRa Radio (such as the SX1272), while disregarding the consumption of other components such as the microcontroller and sensors.

As a result, it is clear that there is ample room for researchers to contribute to these open-source tools or even propose new simulation tools that can provide more comprehensive simulations. Such tools could include support for different LoRa frequencies and more sophisticated energy models that take into account all the relevant components in a LoRa system.

# 5. Channel Propagation Models

Electromagnetic wave propagation, commonly referred to as radio propagation, can be highly unpredictable in urban or forested areas due to the issues caused by the multipath phenomenon [66]. In such situations, interference can lead to fading and result in the radio signal becoming too weak in some areas to be received accurately. In forested areas, multipath is primarily caused by reflections from vegetation, while in urban environments, reflections from buildings and vehicles are the primary cause [67].

Channel propagation models, also known as path loss models, describe the behavior of electromagnetic waves as they travel from a transmitter to a receiver in a wireless communication network [66]. These models can be either empirical, deterministic, or a combination of both [68]. Empirical models, unlike deterministic models, do not rely on terrain data such as elevation or building dimensions, which makes them less computationally intensive [69]. These models are widely used in wireless network simulation tools, as they allow for the estimation and analysis of network coverage based on the effects of propagation parameters, such as the gateway antenna height and frequency [69]. Therefore, to implement a successful network simulation, an accurate propagation model is crucial in establishing reliable communication links and estimating network coverage [15].

### 5.1 LoRaWAN Propagation Models

In a wireless network, the reduction in signal strength between a transmitter and a receiver is referred to as path loss, and it has a direct impact on the probability of packet delivery. Therefore, accurate prediction of path loss associated with a specific LoRa gateway position is crucial for optimizing network coverage and evaluating network performance [15]. Channel propagation models are used to calculate this prediction.

The next sections review the most commonly used propagation models in LoRaWAN network simulations.

#### 5.1.1 Log Distance Path Loss

The Log Distance Path Loss, also referred to as the Log Normal Shadowing or One Slope Model, is an extension of the Free Space model [70]. While the Free Space model only considers an obstacle-free path between the transmitter and receiver, the Log Distance model is used to predict propagation loss in different environments, both indoor and outdoor, taking into account shadowing effects caused by objects such as trees, buildings, and hills [71].

The Log Distance model is a computationally efficient propagation model as it has few parameters to characterize the propagation [72]. The main criterion of the Log Distance model is to consider path loss as a logarithmic function of distance [73]. Therefore, the path loss (PL(d)) can be described by the following equation:

$$PL(d)[dB] = PL(d_0) + 10n \log\left(\frac{d}{d_0}\right) + X_{\sigma}$$
<sup>(1)</sup>

Where  $PL(d_0)$  is the path loss given a reference distance (commonly 1 meter), *n* is the path loss exponent, *d* is the distance between transmitter and receiver; and  $X_{sigma}$  is a zero-mean Gaussian distribution random variable (dB) with a standard deviation  $\sigma$  (dB). This variable is used only when there are shadowing effects in the study environment; otherwise, it assumes value 0 [71].

Another parameter that considerably affects propagation modeling performance is the path loss exponent n [71]. It allows the Log Distance model to adapt to the environment characteristics. The occurrences of floors or walls in the modeled environment, for example, are expressed by it [70]. Table 3 can be used as a reference for commonly used ranges of n values for different scenario types.

Table 3. Common values of n for various scenarios [71].

Scenarios	п
Free space	2
Urban area cellular radio	2.7-3.5
Shadowed urban cellular radio	3-5
Inside a building - LOS	1.6-1.8
Obstructed in building	4-6
Obstructed in Factory	2-3

## 5.1.2 Okumura-Hata

The Okumura-Hata model considers path losses due to shadowing effects such as diffraction, reflection, and scattering of the signal [74]. It is applicable only for the frequency range between 150-1500 MHz and distances between 1-20 km. The Okumura-Hata prediction path loss in urban environments ( $PL_{urban}$ ), in dB, can be described by the equation:

$$PL_{urban} = 69.55 + 26.16 \log f_c - 13.82 \log h_t - \alpha_h + (44.9 - 6.55 \log h_t) \log d$$
(2)

where  $f_c$  is the frequency in MHz,  $h_t$  is transmitter or base station height, ranging from 30 to 200 meters;  $h_r$  is receiver or mobile antenna height, ranging from 1 to 10 meters; d is the distance between transmitter and receiver, and finally  $\alpha_{h_r}$  is the correction factor for receiver antenna height [75]. For small or medium-sized cities, it can be described by the equation:

$$\alpha_{h} = (1.1\log f_c - 0.7)h_r - (1.56\log f_c - 0.8) \tag{3}$$

According to [74], real experiments shown that the path losses for suburban ( $PL_{suburban}$ ) and rural ( $PL_{rural}$ ) environments, can be predicted based on the respective equations:

$$PL_{suburban} = PL_{urban} - 2\left[\log\left(f_c / 28\right)\right]^2 - 5.4 \tag{4}$$

$$PL_{rural} = PL_{urban} - 4.78\log f_c^2 + 18.33\log f_c - 40.98$$
<sup>(5)</sup>

## **5.1.3 Building Penetration Loss**

The Building Penetration Loss is responsible for calculating the path losses that are caused by buildings [76]. The total loss calculation is based on three main components: losses due to external building walls (*EWL*), losses due to internal building walls (*IWL*) and gain in receiving power (*GFH*), proportional to the device's elevation height. Based on these components, the total path loss, in dB, caused by buildings (*PL*<sub>buildings</sub>) for an indoor device is given by:

$$PL_{huildings} = EWL + IWL - GFH \tag{6}$$

The *EWL* component is modeled as a uniform random variable taking values over a certain range of intervals and probability, tabulated in [76]. The range of values that *EWL* can take is intended to model different types of walls (e.g., building material, thickness) [77]. The *IWL*, on the other hand, is calculated by means of the maximum between two values and can be expressed as:

$$IWF = max(Tor1, Tor2) \tag{7}$$

where Tor1 represents the loss due to the walls number and Tor2, represents the loss due to signal penetration into the wall given a uniform distance in the range 0 to 15 meters. Finally, GFH is responsible for expressing the gain in reception that an increase in the end device height produces. The detailed calculation of each of these components can be found in [76].

## **5.1.4 Correlated Shadowing**

Shadowing correlation has an impact on different aspects of wireless networks, such as handover behavior and interference power [77]. Consequently, when conducting network simulations, it is important to incorporate a propagation model that accounts for this correlation. The Correlated Shadowing model considers two types of correlation: firstly, when a transmitter communicates with two nearby devices, it is expected that these devices will experience similar shadowing effects. Secondly, when two transmitters are in close proximity and sending

#### Volume 2 Issue 1|2024| 159

signals to the same node, the receiving device is likely to encounter correlated shadowing in the form of both signals being obstructed by a significant obstacle located between the transmitters and the receiver. A distance-dependent exponential function describes the correlation between the shadowing experienced by nodes i and j as follows:

$$\rho_{i,j}(d_{i,j}) = e^{-d_{i,j}/d_o}$$
(8)

where  $d_{i,j}$  is the distance between nodes *i* and *j* and  $d_0 > 0$  is an adjustable parameter called decorrelation distance, representing the distance at which the shadowing correlation is below the threshold  $e^{-1}$  and therefore shadowing can be considered uncorrelated. The Correlated Shadowing model generates 2D maps that describe the shadowing at various points within a specific region of space, given a transmitter's position. These maps are intended to represent the correlated shadowing effects that occur within the region [77].

# 6. Review of simulation parameters

The utilization of network simulation tools significantly enhances project testing by minimizing expenses, complications, and superfluous use of technologies. Moreover, these tools make it easier to evaluate the performance of different protocols. For effective simulation, precise parameters such as channel model, packet size, and number of nodes are necessary. Despite the transparency of their studies, many research papers that explore network simulations fail to address the matter of reproducibility by other researchers [78].

Ensuring the reproducibility of experiments is crucial for enhancing our understanding of research and serves as a guide for future studies, as well as a means of scientific recognition. It is a fundamental practice for generating new insights. This involves documenting and providing a clear understanding of all the steps and data involved in an experiment, with the aim of producing results that are equivalent or close to those presented by the original authors [79].

Simulation tools provide a wide range of configurable parameters and scenarios that can have a significant impact on performance. Therefore, it is important to address these parameters in the literature. In our study, we examined the literature on network parameters for LoRAWAN and used the findings to compile a list of parameters for review. We expanded upon the work of Gkotsiopoulos et al. [80], who categorized various performance determinants in the literature. Our analysis involved extending their list of parameters, focusing on both simulation and real-world experiments, and examining how the literature addresses each parameter. The list of parameters analyzed is presented in Table 4, with the additional parameters marked with an asterisk (\*), indicating those not present in the previous work.

A reproducibility analysis of the literature on LoRaWAN networks was conducted, covering 17 papers published between 2016 and 2021. These papers addressed various topics, including the scalability of the technology, data monitoring, simulations with collisions, and others. The papers encompassed both real-world and simulated experiments, and were quantified based on the number of essential parameters they presented for reproducing the experiments in a simulation. The observed parameters included: 1. Confirmed type messages (ACK); 2. Signal transmission direction such as Uplink (UL), Downlink (DL) and Direct to Device (D2D); 3. Packet size; 4. Total duration of the experiment; 5. Packet interval; 6. Bandwidth; 7. Channel propagation model; 8. Transmission power; 9. Frequency; 10. Spreading factor; 11. Height of devices, gateways, or antennas; 12. Positions of nodes; and 13. Hardware platforms.

In the subsequent sections, we will explore the usage of these parameters in the literature, their importance, particularly in relation to reproducibility, and the commonalities found in the works that employ them.

	Theoretical		R	eal W	orld						Sin	nulati	on				
	Experiments																
	[81]	[82]	[83]	[84]	[85]	[86]	[58]	[87]	[88]	[89]	[90]	[91]	[92]	[93]	[94]	[69]	[95]
Simulation	Х	Х	Х	Х	Х	Х	NS-3	Radio	NS-3	NS-	NS-	NS-3	Flora	NS-3	NS-	NS-3	NS-3
Tool								Mobile	;	3	3				3		
Confirmed	$\checkmark$	Х	Х	Х	Х	Х	$\checkmark$	Х	Х	Х	Х	$\checkmark$	Х	Х	Х	Х	$\checkmark$
Messages																	
Signal	UL	UL	UL	D2D	UL	UL	UL	UL	UL	UL	UL	UL	UL	UL	UL	UL	UL
Direction	DL					D2D	DL										
Packet	51	Х	22	32	10-50	12	21	Х	21	47	29	28,	20	Х	Х	Х	51
Size (B)			41									14					
Runtime	Х	Х	$1 \mathrm{w}$	30	Х	2 d	100*PI	Х	32000s	×х	Х	Х	10 d	Х	Х	Х	600s
				min		2 months											
Packet Interval	Poisson	0.016s	1h	Х	Х	90 min	600s	Х	400s	Х	100s	600s	30m	10	Х	Х	Х
(PI)	Distribution						6000s		800s		300s			min			
							60000s		1200s		600s						
									1600s								
Bandwidth (KHz)*	125	125	125	125	125	125	125	Х	125	500	Х	125	125	125	Х	125	12
				250													
				500													
Propagation	OH	Х	Х	Х	OH	Х	LD	LR	Х	LD	Х	Х	LD	OH	Х	OH	FSPL
Model*																C-231	
																Cost	
																WI	
TX Power	Х	Х	14	14	$\checkmark$	14	14	20	14	14	Х	14	2 -	14	Х	14	14
(dBm)*													14				
Frequency	Х	433	868	915	868	868	868	915	868	915	868	868	868	923	Х	868	868
(MHz)*				868													
Spreading	7-12	Х	7-12	7-12	7-12	7-12	7-12	Х	7-12	7-12	7-12	7-12	7-12	7-12	7-12	Х	7-12
Factor*																	
GW	Х	2	11-58	Х	25	Х	×	46	Х	Х	15	Х	Х	30	Х	50	15/50
Elevation (m)*																	
Node	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	Х	$\checkmark$										
Distribution*																	

Table 4. Parameters that are explored in the literature.

#### 6.1 Confirmation Messages

This parameter ensures that a packet is delivered by requiring the receiver to send an acknowledgment signal to the sender for the message received. The use of the ACK flag is common in simulation studies to analyze the scalability of LoRa technology. For instance, Bankov et al [81] and Farhad et al [96] evaluate the performance of the LoRaWAN protocol by increasing the density of devices and packets for a single gateway, but the latter also includes unconfirmed type messages and an urban scenario with buildings in their NS-3 simulations. In contrast, Abeele et al [58] consider various parameters along with confirmed and unconfirmed type messages to simulate different scenarios with one or more gateways.

Some authors opt not to use this parameter in their simulations, particularly when experimenting with scenarios that require the fast delivery of critical packets. For example, Priyanta et al [90] simulate a port environment that demands high communication throughput, while Haiahem et al [88] evaluate network scalability performance for air pollution monitoring, which requires a report from each device at specified intervals. Additionally, Haxhibeqiri et al [85] completed its packet collision analysis without utilizing the parameter but may consider it in future work. Finally, Musek et al [86] aim to conserve battery power for its device by utilizing the unconfirmed message mode, and hence does not use the ACK flag.

#### 6.2 Signal Direction

Uplink communication refers to the transmission of data from a node to an application, while downlink communication refers to the transmission of responses from an application to a node. Downlink communication can serve various purposes, such as notifying the device of a specific event or updating an application parameter.

Downlink traffic can negatively impact the performance of the LoRa network, as the gateway cannot receive transmissions while it is sending data. However, Bankov et al [81] and Abeele et al [58] incorporate downlink transmission in their studies to investigate the performance of this parameter in the network. Bankov et al [81] reserve one of the four channels exclusively for the gateway to respond to the devices. On the other hand, Abeele et al [58] indicate that the limited downstream capacity significantly impairs the delivery rate of confirmed upstream message packets. Although increasing the density of the gateway can delay this effect, it cannot entirely eliminate it.

## 6.3 Packet Size

The packet size in LoRa is determined by the regional transmission parameters as well as the proximity to the nearest gateway. According to the LoRa Alliance [96], in the European scenario with a bandwidth of 125kHz, the maximum payload sizes are as follows: 51 bytes for SF10, SF11, and SF12; 115 bytes for SF9; and 242 bytes for SF7 and SF8. In addition, for a scenario with a bandwidth of 250kHz and SF7 the maximum payload is 242 bytes.

The payload sizes in the reviewed works varied between a minimum of 10 bytes and a maximum of 51 bytes, with the European frequency being the most commonly used in the experiments. Bankov et al [81], Ortiz et al [89], and Farhad et al [95], which tested scalability and packet reception mostly in urban and high-density scenarios, used a payload size of 51 bytes. Conversely, Masek et al [86] and Santos Filho et al [91], which focused on long-term and delivery reliability tests, respectively, used smaller payload sizes.

#### 6.4 Duration of Experiments

In our survey, we are considering the duration of an experiment as the exact time spent on real and simulated tests. Based on the analysis of existing literature, it can be observed that there is no standardized approach for determining this parameter. Its selection and application vary based on the specific requirements and limitations outlined in each individual study. In the concrete experiments, as in articles [83] and [86], its duration is based on the characteristics of the collected elements: in the first, one of the experiments has a time of 1 week because it is an application on monitoring radioactivity containers, and this is the standard for garbage collection; in the second, the experiments have 2 days of data collection in the city of Brno, Czech Republic, and 2 months in the long-term test that explores the characteristics of the LoRaWAN channel in the same region. In articles [92] and [94], the total period is based on the experimental repetition method adopted by the authors. While in articles [58,88] and [95], the simulations last in proportion to the packet interval, monitoring periods and a short time of 600 seconds, respectively.

#### 6.5 Packet Interval

We are considering the time duration between packet transmissions for each device. Research studies, such as [91,92] and [93], which focus on constant message transmission for monitoring purposes, have utilized packet intervals not exceeding 30 minutes. In other studies, such as those examining transmission reliability in specific scenarios, found in [58,81] and [90], different packet intervals were employed during experiments. Conversely, articles proposing projects aimed at conserving battery life, such as [83] and [86], feature longer packet sending intervals ranging from 1 hour or more.

#### 6.6 Bandwidth

The LoraWAN protocol offers the option to use channels with a bandwidth of 125kHz, 250kHz, or 500kHz, which is dependent on the region or frequency plan. By increasing the bandwidth, the number of bytes transmitted in the same period is doubled. It is important to note that the design of antennas is influenced by the bandwidth and frequency used [97]. In European experiments, as well as those conducted in Thailand [82] using a frequency of 433MHz and the European configuration, the default bandwidth is 125kHz. In contrast, articles from Brazil [84,89] enable higher bandwidth rates. The former investigates bandwidth intervals in different scenarios, including 500kHz in a forest, 125kHz and 250kHz in suburban and urban areas, and 125kHz for testing mobility in suburban environments. The latter maintains a bandwidth of 500kHz to examine the maximum communication ranges with and without mobility.

## 6.7 Channel Propagation Model

In Section 5, it is elucidated that channel propagation models play a crucial role in capturing the characteristics of electromagnetic waves within wireless communication networks. To provide comprehensive information, Table 5 presents a detailed overview of the propagation models employed in the surveyed papers. The majority of the papers examine urban scenarios with Line-of-Sight (LOS) and/or Non-Line-of-Sight (NLOS) conditions. The study environments mainly consist of Smart Cities or Smart Campuses. The Log Distance Path Loss model is the most commonly used propagation model in LoRaWAN network simulations, but few papers compare the coverage predictions of this model to actual field measurements, with the exception of [89]. Some papers utilize more than one propagation model, as seen in [90,91], and [95]. In addition to the Log Distance (LD) model, they include models such as Building Pen-etration Loss and Correlated Shadowing Propagation Loss that account for the shadowing effects caused by objects such as buildings and vehicles on electromagnetic waves.

	Year	Propagation model	Freq (MHz)	Environment	Propagation
[81]	2016	Okumura-Hata	868	Smart City	-
[85]	2017	Okumura-Hata	868	Smart City	-
[58]	2017	Log Distance	868	Smart City	LOS
[87]	2021	Longley-Rice (LR)	915	Smart City	LOS, NLOS
[88]	2020	-	868	Smart City	NLOS
[89]	2020	Log Distance	915	Smart Campus	LOS
[90]	2019	Log Distance + Building Propagation + Correlated Shadowing	868	Smart City	LOS, NLOS
[91]	2020	Log Distance + Building Propagation	868	Smart City	LOS, NLOS
[92]	2019	Log Distance	868	Smart City	LOS
[93]	2020	Okumura-Hata	AS923	Smart City	LOS
[94]	2019	-	-	Smart Campus	LOS, NLOS
[69]	2019	Okumura-Hata COST-231 Hata (C-231) COST WI	868	Smart City	NLOS
[96]	2019	Log Distance + Building Propagation	868	Smart City	LOS, NLOS

Table 5. Propagation model details

Out of the papers surveyed, only four of them [81,85,93], and [69], used the Okumura-Hata (OH) propagation model. However, only [85] and [69] validated this model in the field. Therefore, it can be concluded that validating the results of propagation model predictions and achieving accuracy in the analysis of network performance in terms of packet delivery is still a challenge in some research works that study LoRa coverage analysis through simulations, as comparison with real data is not always performed.

#### 6.8 Transmission Power

In Europe, the maximum transmission power allowed for uplink transmission is 14dBm [98], and most of the surveyed works use this default value. However, in [86], the authors point out that this value was chosen to allow for longer communication ranges. In [87], Ortega et al. study an urban scenario in Colombia (using the US902-928MHz frequency band) and choose a transmission power value of 20dBm to avoid noise in the incoming signal. Another approach to transmission power can be found in [92], which uses Adaptive Data Rate (ADR) with a range of transmission power values between 2dBm and 14dBm in order to minimize energy consumption.

## 6.9 Frequency

The choice of allowable transmission power, bandwidth, and antenna radiation intensity in a LoRa network is determined by the regulations of the country or region, as well as the appropriate frequency plan. Hence, the frequency value is a crucial factor in experimentation. Further information on each frequency plan and Volume 2 Issue 1/2024/163 Computer Networks and Communications

corresponding countries can be found on the LoRa Alliance website. Among the reviewed articles, four different frequency values were used, including 433 MHz (EU433 in Asia), 868 MHz (European standard), 915 MHz (Latin America and Australia), and 923 MHz (AS923 in Asia). Interestingly, some experiments conducted in in the Americas [84] and Africa [92] used the 868 MHz frequency.

It is essential to consider the distinct characteristics of LoRaWAN supported frequencies. Comparing the two primary frequencies for LoRaWAN in Europe, 433 MHz and 868 MHz, as defined by the LoRa Alliance standard [97], reveals significant differences. The former requires a duty cycle (channel occupancy rate) of less than 10%, meaning that EU433 devices can only transmit data for 10% of the total time, with a maximum EIRP (Isotropic Effective Radiated Power) standard of 12 dBm. In contrast, for the EU868 frequency, the duty cycle is even stricter, with a lower limit of 1% per day, and its maximum EIRP is 16 dBm, allowing for greater transmission power. While the LoRaWAN AS923 frequency, used in Asia, shares characteristics with EU868, the AU915 frequency, utilized in Latin America and Australia, has no duty cycle limit, and its maximum EIRP is 30 dBm.

## 6.10 Spreading Factor Characteristics

The Spreading Factor is a crucial parameter in LoRa technology that controls the transmission speed. Smaller SFs result in faster transmissions with a higher data rate, while larger SFs lead to slower transmissions with a lower data rate. Since signals modulated with different spreading factors are orthogonal, there is no interference between them when sent simultaneously on the same channel frequency. Therefore, SF is an important parameter used in congestion control in a network, and it affects the data rate, time of a message in the air (ToA), battery expenditure, and receiver sensitivity. LoRa technology has a total of 6 spreading factors, ranging from SF7 to SF12 [99].

The selection of spreading factors is of great importance in determining the performance of a LoRa network. Transmissions with the same SF may experience intra-SF interference when they overlap in time, while transmissions with different SFs may suffer from inter-SF interference. Additionally, higher SF values can reduce network capacity due to longer data transfer times, whereas multiple SFs can increase capacity significantly due to parallel transmissions that are mostly orthogonal to each other.

In the reviewed papers, the SF values in the devices are adapted to the specific problem being investigated. Smaller SFs are used to reduce battery consumption, higher SFs are used to achieve long ranges, and varying intervals are used to find the best performance.

Van den Abeele et al. [58] used an error model to test three types of allocation: random, fixed (same SF for all devices), and PER (smallest SF whose error rate per packet falls below a threshold for each device). PER had the best results and was used in the final design.

Santos Filho et al. [91] conducted a performance evaluation by testing three different strategies to simulate an industrial scenario, involving nodes with varying priority levels: higher (alarms) and lower (regular) priority. The first strategy, referred to as the basic strategy, assigns the lowest SF based on the received power at the receiver, with a transmit power of 14 dBm. The second strategy, known as the shift strategy, allocates each node to the SF one level higher than the one assigned in the basic strategy. However, the authors found the SF reservation strategy, where alarm nodes are assigned to the highest SF among them in the basic strategy while regular nodes cannot be assigned to this specific SF, to be an inadequate solution.

In the street lighting system, presented in [92], the Adaptive Data Rate (ADR) based on Signal-to-Noise Ratio (SNR) is used. For a large number of gateways, devices use smaller SF, which reduces power consumption and extends network lifetime.

Mnguni et al. [94] present gateway positioning algorithms that assign the SF after simulating the network five times and calculating the signal strength between the device and gateway.

#### 6.11 Gateway Position and Elevation

The placement of gateways also impacts the performance of LoRaWAN experiments. The literature suggests that locations with fewer obstacles, lower signal penetration loss, and support for multiple SFs, especially those with lower ToA values, have shown better results. Several studies, such as [69,83,85,87,90,94], and [95], have been conducted in urban areas, where gateway heights are chosen to mitigate signal interference. The gateways are typically placed on top of or above buildings, with minimum and maximum heights of 15 m and 50 m, respectively. These limits have been used in [95], which demonstrated that increasing the antenna height significantly improves the success rate of receiving packets.

In other studies, such as [58,91], and [93], the placement of gateways is determined differently. In the first two studies, the gateways are placed inside a circular area with the devices, and their coordinates are fixed based

on the number of simulated gateways. In [93], the height of the gateways varies from 30 m to 200 m above the ground level, according to the limits of the propagation model used.

In contrast, [82] sets the same height for the device and the gateway in the first experiment to test the rate of lost packets by changing the distance.

## 6.12 Node Distribution

The distribution of nodes across a geographical area is another important factor in evaluating the performance of a LoRaWAN network. The number and location of gateways used in the experiment have a significant impact on the network's ability to transmit and receive packets. Overcrowding in a particular area can decrease the network's performance, while placing nodes near gateways can improve it.

In experiments that simulate scenarios with a single gateway, such as those in [81,85,88,90,94,95], the performance is tested with a varying number of devices, including up to 10,000 devices in some cases, which can overcrowd the area and negatively affect the performance of the network.

On the other hand, experiments in [82,83,84,86,87,89] and [93], which are either real experiments or simulations based on existing telemetry systems, use a smaller number of nodes fixed in specific positions to measure targeted metrics.

In some studies, such as [58,91], and [92], multiple gateways are used, and their positions are varied to evaluate network performance. Increasing the gateway density can mitigate the negative impact of overcrowding, but cannot completely eliminate it.

## 6.13 Hardware Platforms

Real-world experiments were conducted to evaluate the performance of LoRaWAN Networks in papers such as [82,83,84], and [86]. These experiments utilized different types of hardware, showcasing a wide range of options for end devices, LoRa radio modules, and gateways. Table 6 presents a description of the primary hardware used in the evaluation of LoRaWAN Networks.

Hardware	[82]	[83]	[84]	[86]
End Device	Arduino Uno	ATSAML21G18B	Arduino Uno	Designed
	Libelium Lora			development kit
LoRa radio module	Libelium Lora	RFM95W-868S2	Dragino RF96	Microchip
		ANT-868-JJB-ST	Monopole 7dBi	RN2483
Gateway	NodeMCUV2	MTCDT-210A	-	-

Table 6. Hardware description of LoraWAN work using real world experiment.

In [82] and [84], the Arduino Uno microcontroller was chosen. In [83] and [86], the ATSAML21G18B micro-controller and a board developed by the authors themselves with the built-in Microchip RN2483 LoRa radio were utilized, respectively. Regarding other radio modules, the works [82,83], and [84] employed the Libelium LoRa module (433MHz), RFM95W-868S2, and Dragino RF96, respectively. The gateway used in [82] and [83] consisted of the NodeMCUV2 microcontroller and the Libelium LoRa radio module, while the MultiConnect Conduit model MTCDT-210A was used. In [84], Dragino RF96 kits were used for device-to-device communication, and B-L072Z-LRWAN1 LoRA/Sigfox kits with an onboard Semtech SX1276 transceiver chip were used for device-to-gateway communication.

# 7. Conclusion and Future Research Directions

This survey provides a comprehensive overview of LoRa and LoRaWAN technologies, with a particular focus on simulation tools used in research papers and the various parameters employed for simulating these networks. The survey also discusses challenges and opportunities that exist within this research area.

A key contribution of this work lies in the detailed and critical analysis of the significance of LoRaWAN parameters in the existing literature. To achieve this, both simulation-based and real-world studies were reviewed, elucidating the key characteristics of the technology as described by these works. This analysis offers valuable insights into how authors approach LoRaWAN experiments and their perspectives on the essential parameters for future studies to reproduce.

In Section 4, it was demonstrated that while several options for LoRa and LoRaWAN simulation tools already exist, the majority of them primarily focus on reproducing European (868 MHz) simulation scenarios, neglecting implementations for other regions. Additionally, the absence of complex energy models in these tools was observed, as most of them only consider the energy consumption of the LoRa Radio, disregarding the microcontroller and sensor consumption.

Another challenge arises from the limitations of certain tools in validating predicted propagation models in LoRa networks using real measurements, which compromises the accuracy of network performance assessment, particularly in terms of packet delivery. Furthermore, most of the available simulation tools lack robust class and parameter settings for the Network Server. Consequently, there are ample opportunities for contributing to the improvement of these simulation tools by addressing these limitations.

# Acknowledgments

This work was funded by Konker Labs, Brazil grant #34-P-04058-2018 and by the Brazilian National Council for Scientific and Technological Development (CNPq) grant #308840/2020-8.

# **Conflict of Interest**

There is no conflict of interest for this study.

# References

- Madakam, S.; Ramaswamy, R.; Tripathi, S. Internet of Things (IoT): A literature review. J. Comput. Commun. 2015, 3, 164–173, https://doi.org/10.4236/jcc.2015.35021.
- [2] Mukhopadhyay, S.C.; Suryadevara, N.K. Internet of Things: Challenges and Opportunities. In Internet of Things. Smart Sensors, Measurement and Instrumentation, Springer: Cham, Switzerland, https://doi.org/10. 1007/978-3-319-04223-7\_1.
- [3] Mekki, K.; Bajic, E.; Chaxel, F.; Meyer, F. A comparative study of LPWAN technologies for large-scale I oT deployment. *ICT Express* 2019, 5, 1–7, https://doi.org/10.1016/j.icte.2017.12.005.
- [4] Bahashwan, A.A.; Anbar, M.; Abdullah, N.; Al-Hadhrami, T.; Hanshi, S.M. Review on Common IoT Co mmunication Technologies for Both Long-Range Network (LPWAN) and Short-Range Network. In Adva nces on Smart and Soft Computing. Advances in Intelligent Systems and Computing, Springer: Singapore, Singapore, https://doi.org/10.1007/978-981-15-6048-4 30.
- [5] Catherwood, P.; Rafferty, J.; McComb, S.; McLaughlin, J. LPWAN Wearable intelligent healthcare monitoring for heart failure prevention. In Proceedings of British HCI Conference 2018, Belfast, Ireland, 2–6 July 2018.
- [6] Grunwald, A.; Schaarschmidt, M.; Westerkamp, C. LoRaWAN in a rural context: Use cases and opportuni ties for agricultural businesses. In Proceedings of Mobile Communication - Technologies and Applications; 24. ITG-Symposium, Osnabrueck, Germany, 15–16 May 2019.
- [7] Civelek, C. Low power wide area network (lpwan) and internet of things adaptation in agricultural machin ery. *Sch. J. Agric. Vet. Sci.* **2017**, *4*, 18–23, https://doi.org/10.21276/sjavs.2017.4.1.4.
- [8] Tarjan, L.; Tejić, B.; Dragičević, D.; Ostojić, G.; Đukić, N. An alternative communication possibility duri ng a natural disaster by using lowpower long-range RF modules. In Proceedings of 2018 17th International Symposium INFOTEH-JAHORINA (INFOTEH), East Sarajevo, Bosnia and Herzegovina, 21–23 March 2018, https://doi.org/10.1109/INFOTEH.2018.8345527.
- [9] Low-power wide area network (lpwan): Overview. Available online: https://tools.ietf.org/html/rfc8376. (a ccessed on 30 April 2024)
- [10] Ts001-1.0.4 lorawan l2 1.0.4 specification. Available online: https://resources.lora-alliance.org/technical-s pecifications/ts001-1-0-4-lorawan-l2-1-0-4-specification. (accessed on 30 April 2024)
- [11] Technical specification group radio access network study on provision of low-cost machine-type communi cations (mtc) user equipments (ues) based on lte. Available online: https://portal.3gpp.org/desktopmodules /Specifications/SpecificationDetails.aspx?specificationId=2578. (accessed on 11 February 2023)
- [12] Dangi, R.; Lalwani, P.; Choudhary, G.; You, I.; Pau, G. Study and Investigation on 5G Technology: A Sys tematic Review. Sensors 2021, 22, 26, https://doi.org/10.3390/s22010026.

- [13] Guo, F.; Yu, F.R.; Zhang, H.; Li, X.; Ji, H.; Leung, V.C.M. Enabling Massive IoT Toward 6G: A Compre hensive Survey. *IEEE Internet Things J.* 2021, *8*, 11891–11915, https://doi.org/10.1109/JIOT.2021.30636 86.
- [14] Rappaport, T.S.; Xing, Y.; Kanhere, O.; Ju, S.; Madanayake, A.; Mandal, S.; Alkhateeb, A.; Trichopoulos, G.C. Wireless Communications and Applications Above 100 GHz: Opportunities and Challenges for 6G a nd Beyond. *IEEE Access* 2019, 7, 78729–78757, https://doi.org/10.1109/access.2019.2921522.
- [15] Petrariu, A.I.; Mutescu, P.-M.; Coca, E.; Lavric, A.; Petrariu, A.I.; I., A.; Mutescu, P.-M.; M., P.; Coca, E.; Lavric, A. A Study on LoRa Signal Propagation Models in Urban Environments for Large-scale Networks Deployment. *Adv. Electr. Comput. Eng.* 2021, 21, 61–68, https://doi.org/10.4316/aece.2021.04007.
- [16] Alenezi, M.; Chai, K.K.; Chen, Y.; Jimaa, S. Ultra-dense LoRaWAN: Reviews and challenges. IET Comm un. 2020, 14, 1361–1371, https://doi.org/10.1049/iet-com.2018.6128.
- [17] Ayoub, W.; Samhat, A.E.; Nouvel, F.; Mroue, M.; Prévotet, J.C. Internet of mobile things: Overview of lo rawan, dash7, and nb-iot in lpwans standards and supported mobility. *IEEE Commun. Surv. Tutor.* 2019, 2 1, 1561–1581, https://doi.org/10.1109/COMST.2018.2877382.
- [18] de Carvalho Silva, J.; Rodrigues, J.J.; Alberti, A.M.; Solic, P.; Aquino, A.L. LoRaWAN—A low power W AN protocol for Internet of Things: A review and opportunities. In Proceedings of 2017 2nd International multidisciplinary conference on computer and energy science (SpliTech), Split, Croatia, 12–14 July 2017.
- [19] Aldahdouh, K.A.; Darabkh, K.A.; Al-Sit, W. A survey of 5G emerging wireless technologies featuring Lo RaWAN, Sigfox, NB-IoT and LTE-M. In Proceedings of 2019 International conference on wireless comm unications signal processing and networking (WiSPNET), Chennai, India, 21–23 March 2019, https://doi.o rg/10.1109/WiSPNET45539.2019.9032817.
- [20] Triantafyllou, A.; Sarigiannidis, P.; Lagkas, T.D. Network protocols, schemes, and mechanisms for interne t of things (iot): Features, open challenges, and trends. *Wirel. Commun. Mob. Comput.* 2018, 5349894, htt ps://doi.org/10.1155/2018/5349894.
- [21] Lavric, A.; Popa, V. Internet of things and LoRa<sup>™</sup> low-power wide-area networks: a survey. In Proceedin gs of 2017 International Symposium on Signals, Circuits and Systems (ISSCS), Iasi, Romania, 13–14 July 2017, https://doi.org/10.1109/ISSCS.2017.8034915.
- [22] Saari, M.; bin Baharudin, A.M.; Sillberg, P.; Hyrynsalmi, S.; Yan, W. LoRa—A survey of recent research t rends. In Proceedings of 2018 41st international convention on information and communication technolog y, electronics and microelectronics (MIPRO), Opatija, Croatia, 21–25 May 2018, https://doi.org/10.23919/ MIPRO.2018.8400161.
- [23] Sarker, V.K.; Queralta, J.P.; Gia, T.N.; Tenhunen, H.; Westerlund, T. A survey on LoRa for IoT: Integrati ng edge computing. In Proceedings of 2019 Fourth International Conference on Fog and Mobile Edge Co mputing (FMEC), Rome, Italy, 10–13 June 2019, https://doi.org/10.1109/FMEC.2019.8795313.
- [24] Haxhibeqiri, J.; De Poorter, E.; Moerman, I.; Hoebeke, J. A Survey of LoRaWAN for IoT: From Technolo gy to Application. *Sensors* 2018, 18, 3995, https://doi.org/10.3390/s18113995.
- [25] Khutsoane, O.; Isong, B.; Abu-Mahfouz, A.M. IoT devices and applications based on LoRa/LoRaWAN. I n Proceedings of IECON 2017-43rd Annual Conference of the IEEE Industrial Electronics Society, Beijin g, China, 29 October–1 November 2017, https://doi.org/10.1109/IECON.2017.8217061.
- [26] Adelantado, F.; Vilajosana, X.; Tuset-Peiro, P.; Martinez, B.; Melia-Segui, J.; Watteyne, T. Understanding the limits of LoRaWAN. *IEEE Commun. Mag.* 2017, 55, 34–40, https://doi.org/10.1109/MCOM.2017.16 00613.
- [27] Ertürk, M.A.; Aydın, M.A.; Büyükakkaşlar, M.T.; Evirgen, H. A Survey on LoRaWAN Architecture, Prot ocol and Technologies. *Future Internet* 2019, 11, 216. https://doi.org/10.3390/fi11100216.
- [28] Marais, J.M.; Abu-Mahfouz, A.M.; Hancke, G.P. A Survey on the Viability of Confirmed Traffic in a LoR aWAN. *IEEE Access* 2020, 8, 9296–9311, https://doi.org/10.1109/access.2020.2964909.
- [29] Banti, K.; Karampelia, I.; Dimakis, T.; Boulogeorgos, A.-A.A.; Kyriakidis, T.; Louta, M. LoRaWAN Com munication Protocols: A Comprehensive Survey under an Energy Efficiency Perspective. *Telecom* 2022, 3, 322–357, https://doi.org/10.3390/telecom3020018.
- [30] Jouhari, M.; Saeed, N.; Alouini, M.-S.; Amhoud, E.M. A Survey on Scalable LoRaWAN for Massive IoT: Recent Advances, Potentials, and Challenges. *IEEE Commun. Surv. Tutorials* 2023, 25, 1841–1876, https: //doi.org/10.1109/comst.2023.3274934.
- [31] Ferrari, P.; Sisinni, E.; Carvalho, D.F.; Depari, A.; Signoretti, G.; Silva, M.; Silva, D. On the use of LoRa WAN for the Internet of Intelligent Vehicles in Smart City scenarios. In Proceedings of 2020 IEEE Sensor s Applications Symposium (SAS), Kuala Lumpur, Malaysia, 9–11 March 2020, https://doi.org/10.1109/S AS48726.2020.9220069.

- [32] Klaina, H.; Guembe, I.P.; Lopez-Iturri, P.; Astrain, J.J.; Azpilicueta, L.; Aghzout, O.; Alejos, A.V.; Falcon e, F. Aggregator to Electric Vehicle LoRaWAN Based Communication Analysis in Vehicle-to-Grid Syste ms in Smart Cities. *IEEE Access* 2020, *8*, 124688–124701, https://doi.org/10.1109/ACCESS.2020.300759 7.
- [33] Basford, P.J.; Bulot, F.M.J.; Apetroaie-Cristea, M.; Cox, S.J.; Ossont, S.J. LoRaWAN for Smart City IoT Deployments: A Long Term Evaluation. Sensors 2020, 20, 648, https://doi.org/10.3390/s20030648.
- [34] Yu, F.; Zhu, Z.; Fan, Z. Study on the feasibility of LoRaWAN for smart city applications. In Proceedings o f 2017 IEEE 13th International conference on wireless and mobile computing, networking and communica tions (WiMob), Rome, Italy, 9–11 October 2017, https://doi.org/10.1109/WiMOB.2017.8115748.
- [35] Almuhaya, M.A.M.; Jabbar, W.A.; Sulaiman, N.; Abdulmalek, S. A Survey on LoRaWAN Technology: R ecent Trends, Opportunities, Simulation Tools and Future Directions. *Electronics* 2022, 11, 164, https://do i.org/10.3390/electronics11010164.
- [36] Bouras, C., Gkamas, A., Katsampiris Salgado, S.A., Kokkinos, V. (2020). Comparison of LoRa Simulatio n Environments. In Advances on Broad-Band Wireless Computing, Communication and Applications. B WCCA 2019. Lecture Notes in Networks and Systems, Springer: Cham, Switzerland, https://doi.org/10.10 07/978-3-030-33506-9\_33.
- [37] da Silva, J.C.; Flor, D.D.L.; de Sousa Junior, V.A.; Bezerra, N.S.; de Medeiros, A.A. A survey of LoRaW AN simulation tools in ns-3. J. Commun. Inf. Syst. 2021, 36, 17–30, https://doi.org/10.14209/jcis.2021.2.
- [38] Raza, U.; Kulkarni, P.; Sooriyabandara, M. Low Power Wide Area Networks: An Overview. IEEE Comm un. Surv. Tutor. 2017, 19, 855–873, https://doi.org/10.1109/comst.2017.2652320.
- [39] Milarokostas, C.; Tsolkas, D.; Passas, N.; Merakos, L. A Comprehensive Study on LPWANs With a Focus on the Potential of LoRa/LoRaWAN Systems. *IEEE Commun. Surv. Tutor.* 2023, 25, 825–867, https://doi. org/10.1109/COMST.2022.3229846.
- [40] Lpwan market report 2018-2023. Available online: https://iot-analytics.com/product/lpwan-market-report-2018-2023/. (accessed on 3 November 2019)
- [41] Mekki, K.; Bajic, E.; Chaxel, F.; Meyer, F. Overview of cellular LPWAN technologies for IoT deploymen t: Sigfox, LoRaWAN, and NB-IoT. In Proceedings of 2018 ieee international conference on pervasive com puting and communications workshops (percom workshops), Athens, Greece, 19–23 March 2018, https://d oi.org/10.1109/PERCOMW.2018.8480255.
- [42] Alliance, L. What is lorawan? Available online: https://lora-alliance.org/about-lorawan/. (accessed on 11 February 2023).
- [43] Hornbuckle, C.A. Fractional-n synthesized chirp generator. United States Patent US7791415B2, Semtech Corp, May 2007.
- [44] Li, C.; Ren, Y.; Tong, S.; Siam, S.I.; Zhang, M.; Wang, J.; Liu, Y.; Cao, Z. ChirpTransformer: Versatile L oRa Encoding for Low-power Wide-area IoT. In Proceedings of the 22nd Annual International Conference on Mobile Systems, Applications and Services, Tokyo, Japan, 3–7 June 2024, https://doi.org/10.1145/364 3832.3661861.
- [45] Knight, M.; Seeber, B. Decoding LoRa: Realizing a modern LPWAN with SDR. In Proceedings of the GN U Radio Conference, Boulder, CO, USA, 12–16 September 2016.
- [46] Harris, N.R.; Curry, J. Development and range testing of a LoRaWAN system in an urban environment. In t. J. Electron. Commun. Eng. 2018, 12, 47–55, https://doi.org/10.5281/zenodo.1315517.
- [47] Henriksson, R. Indoor positioning in LoRaWAN networks. Master's thesis, Chalmers University of Techn ology: Gothenburg, Sweden, 2016.
- [48] Chirpstack network server. Available online: https://www.chirpstack.io/network-server/overview/. (access ed on 3 November 2019)
- [49] Ursalink embedded lorawan network server. Available online: https://www.ursalink.com/en/blog/embedde d-lorawan-network-server/. (accessed on 3 November 2019)
- [50] Mdhaffar, A.; Chaari, T.; Larbi, K.; Jmaiel, M.; Freisleben, B. IoT-based health monitoring via LoRaWA N. In Proceedings of IEEE EUROCON 2017-17th international conference on smart technologies, Ohrid, Macedonia, 6–8 July 2017, https://doi.org/10.1109/EUROCON.2017.8011165.
- [51] Hybrid network management system for massive iot. Available online: https://www.loriot.io/about.html. (a ccessed on 22 February 2023)
- [52] Tayebi, A.; Veltri, L.; Zanichelli, F.; Caselli, S. Interworking between LoRaWAN and non-LoRa IoT syste ms. In Proceedings of 2022 IEEE International Conference on Pervasive Computing and Communications Workshops and other Affiliated Events (PerCom Workshops), Pisa, Italy, 21–25 March 2022, https://doi.o rg/10.1109/PerComWorkshops53856.2022.9767494.
- [53] Mqtt the standard for iot messaging. Available online: https://mqtt.org/. (accessed on 11 February 2023).

- [54] Mishra, B.; Kertesz, A. The Use of MQTT in M2M and IoT Systems: A Survey. *IEEE Access* **2020**, *8*, 201 071–201086, https://doi.org/10.1109/ACCESS.2020.3035849.
- [55] You, I.; Kwon, S.; Choudhary, G.; Sharma, V.; Seo, J.T. An Enhanced LoRaWAN Security Protocol for P rivacy Preservation in IoT with a Case Study on a Smart Factory-Enabled Parking System. *Sensors* 201 8, 18, 1888, https://doi.org/10.3390/s18061888.
- [56] What is ns-3? Available online: https://www.nsnam.org/about/. (accessed on 22 February 2023).
- [57] Reynders, B.; Wang, Q.; Pollin, S. A LoRaWAN module for ns-3: Implementation and evaluation. In Proc eedings of the 2018 Workshop on ns-3, Surathkal, India, 13–14 June 2018, https://doi.org/10.1145/319990 2.3199913.
- [58] Van den Abeele, F.; Haxhibeqiri, J.; Moerman, I.; Hoebeke, J. Scalability Analysis of Large-Scale LoRaW AN Networks in ns-3. *IEEE Internet Things J.* 2017, *4*, 2186–2198, https://doi.org/10.1109/jiot.2017.2768 498.
- [59] To, T.H.; Duda, A. Simulation of LoRa in NS-3: Improving LoRa performance with CSMA. In Proceedin gs of 2018 IEEE International Conference on Communications (ICC), Kansas City, MO, USA, 20–24 May 2018, https://doi.org/10.1109/ICC.2018.8422800.
- [60] Magrin, D.; Centenaro, M.; Vangelista, L. Performance evaluation of LoRa networks in a smart city scenar io. In Proceedings of the 2017 IEEE International Conference on Communications (ICC), Paris, France, 21 –25 May 2017, https://doi.org/10.1109/ICC.2017.7996384.
- [61] Magrin, D.; Capuzzo, M.; Zanella, A. A Thorough Study of LoRaWAN Performance Under Different Para meter Settings. *IEEE Internet Things J.* 2019, 7, 116–127, https://doi.org/10.1109/jiot.2019.2946487.
- [62] Capuzzo, M.; Magrin, D.; Zanella, A. Confirmed traffic in LoRaWAN: Pitfalls and countermeasures. In Pr occeedings of 2018 17th Annual Mediterranean Ad Hoc Networking Workshop (Med-Hoc-Net), Capri, Ital y, 20–22 June 2018, https://doi.org/10.23919/MedHocNet.2018.8407095.
- [63] Slabicki, M.; Premsankar, G.; Di Francesco, M. Adaptive configuration of LoRa networks for dense IoT d eployments. In Proceedings of NOMS 2018-2018 IEEE/IFIP Network Operations and Management Symp osium, Taipei, Taiwan, 23–27 April 2018, https://doi.org/10.1109/NOMS.2018.8406255.
- [64] Bor, M.C.; Roedig, U.; Voigt, T.; Alonso, J.M. Do LoRa low-power wide-area networks scale? In Proceed ings of the 19th ACM international conference on modeling, analysis and simulation of wireless and mobil e systems, Malta, Malta, 13–17 November 2016, https://doi.org/10.1145/2988287.2989163.
- [65] Spreading factors. Available online: https://www.thethingsnetwork.org/docs/lorawan/spreading-factors/. (a ccessed on 22 February 2023).
- [66] Ortiz, F.M.; Almeida, T.; Ferreira, A.E.L.A.; Costa, L.H.M.K. Caracterização de desempenho de uma rede lora em ambientes urbanos: Simulação vs. prática. In Anais do III Workshop de Computação Urbana, http s://doi.org/10.5753/courb.2019.7476.
- [67] EletronicNotes, Multipath Propagation. Available online: https://www.electronics-notes.com/articles/anten nas-propagation/propagation-overview/multipath-propagation.php. (accessed on 22 February 2023)
- [68] Fernández, J.R.; Quispe, M.; Kemper, G.; Samaniego, J.; Díaz, D. Adjustments of log-distance path loss m odel for digital television in lima. In Proceedings of Anais de 30th Simpósio Brasileiro de Telecomunicaçõ es, Brasília, Brazil, 13–16 September 2012, pp.13–16.
- [69] Harinda, E.; Hosseinzadeh, S.; Larijani, H.; Gibson, R.M. Comparative performance analysis of empirical propagation models for lorawan 868mhz in an urban scenario. In Proceedings of 2019 IEEE 5th World For um on Internet of Things (WF-IoT), Limerick, Ireland, 15–18 April 2019, https://doi.org/10.1109/WF-IoT. 2019.8767245.
- [70] Bertoldo, S.; Paredes, M.; Carosso, L.; Allegretti, M.; Savi, P. Empirical indoor propagation models for Lo Ra radio link in an office environment. In Proceedings of 2019 13th European Conference on Antennas an d Propagation (EuCAP), Krakow, Poland, 31 March–5 April 2019, pp. 1–5.
- [71] M. Viswanathan, Log Distance Path Loss or Log Normal Shadowing Model. Available online: https://www.gaussianwaves.com/2013/09/log-distance-path-loss-or-log-normal-shadowing-model/. (accessed on 11 February 2023)
- [72] Hosseinzadeh, S.; Almoathen, M.; Larijani, H.; Curtis, K. A Neural Network Propagation Model for LoRa WAN and Critical Analysis with Real-World Measurements. *Big Data Cogn. Comput.* 2017, *1*, 7, https://d oi.org/10.3390/bdcc1010007.
- [73] El Chall, R.; Lahoud, S.; El Helou, M. LoRaWAN Network: Radio Propagation Models and Performance Evaluation in Various Environments in Lebanon. *IEEE Internet Things J.* 2019, 6, 2366–2378, https://doi. org/10.1109/jiot.2019.2906838.
- [74] Rofi, A.S.M.; Habaebi, M.H; Islam, M.R.; Basahel, A. LoRa Channel Propagation Modelling using Artific ial Neural Network. In Proceedings of 2021 8th International Conference on Computer and Communicatio

Volume 2 Issue 1|2024| 169

n Engineering (ICCCE), Kuala Lumpur, Malaysia, 22–23 June 2021, https://doi.org/10.1109/ICCCE50029. 2021.9467234.

- [75] Dalveren, Y.; Kara, A. Performance evaluation of empirical path loss models for a linear wireless sensor n etwork deployment in suburban and rural environments. *Hittite J. Sci. Eng.* 2020, 7, 313–320, https://doi.o rg/10.17350/HJSE19030000200.
- [76] Magrin, D. Network level performances of a LoRa system. Master Thesis, University of Padua: Padua, Italy, December 2016.
- [77] Capuzzo, M. Reliable LoRaWAN links: performance analysis. University of Padua: Padua, Italy, March 2019.
- [78] Uhrmacher, A.M.; Brailsford, S.; Liu, J.; Rabe, M.; Tolk, A. Panel—Reproducible research in discrete eve nt simulation—A must or rather a maybe? In Proceedings of 2016 Winter Simulation Conference (WSC), Washington, DC, USA, 11–14 December 2016, https://doi.org/10.1109/WSC.2016.7822185.
- [79] Samuel, S.; König-Ries, B. Understanding experiments and research practices for reproducibility: an explo ratory study. *PeerJ* 2021, 9, e11140, https://doi.org/10.7717/peerj.11140.
- [80] Gkotsiopoulos, P.; Zorbas, D.; Douligeris, C. Performance Determinants in LoRa Networks: A Literature Review. *IEEE Commun. Surv. Tutorials* 2021, 23, 1721–1758, https://doi.org/10.1109/comst.2021.309040 9.
- [81] Bankov, D.; Khorov, E.; Lyakhov, A. On the limits of LoRaWAN channel access. In Proceedings of 2016 International conference on engineering and telecommunication (EnT), Moscow, Russia, 29–30 November 2016, https://doi.org/10.1109/EnT.2016.011.
- [82] Vatcharatiansakul, N.; Tuwanut, P.; Pornavalai, C. Experimental performance evaluation of LoRaWAN: A case study in Bangkok. In Proceedings of 2017 14th International Joint Conference on Computer Science and Software Engineering (JCSSE), NakhonSiThammarat, Thailand, 12–14 July 2017, https://doi.org/10.1 109/JCSSE.2017.8025948.
- [83] Manzano, L.G.; Boukabache, H.; Danzeca, S.; Heracleous, N.; Murtas, F.; Perrin, D.; Silari, M. An IoT Lo RaWAN network for environmental radiation monitoring. *IEEE Trans. Instrum. Meas.* 2021, 70, 1–12, htt ps://doi.org/10.1109/TIM.2021.3089776.
- [84] Ferreira, A.E.; Ortiz, F.M.; Costa, L.H.M.; Foubert, B.; Amadou, I.; Mitton, N. A study of the LoRa signal propagation in forest, urban, and suburban environments. *Ann. Telecomm.* 2020, 75, 333–351, https://doi. org/10.1007/s12243-020-00789-w.
- [85] Haxhibeqiri, J.; Abeele, F.V.D.; Moerman, I.; Hoebeke, J. LoRa Scalability: A Simulation Model Based o n Interference Measurements. *Sensors* 2017, 17, 1193, https://doi.org/10.3390/s17061193.
- [86] Masek, P.; Stusek, M.; Svertoka, E.; Pospisil, J.; Burget, R.; Lohan, E.S.; Marghescu, I.; Hosek, J.; Ometo v, A. Measurements of LoRaWAN Technology in Urban Scenarios: A Data Descriptor. Data 2021, 6, 62, https://doi.org/10.3390/data6060062.
- [87] Ortega-Gonzalez, L.; Acosta-Coll, M.; Piñeres-Espitia, G.; Butt, S.A. Communication protocols evaluation for a wireless rainfall monitoring network in an urban area. *Heliyon* 2021, 7, e07353, https://doi.org/10.10 16/j.heliyon.2021.e07353.
- [88] Haiahem, R.; Minet, P.; Boumerdassi, S.; Saidane, L.A. Collision-Free Transmissions in an IoT Monitorin g Application Based on LoRaWAN. *Sensors* 2020, 20, 4053, https://doi.org/10.3390/s20144053.
- [89] Ortiz, F.M.; de Almeida, T.T.; Ferreira, A.E.; Costa, L.H.M. Experimental vs. simulation analysis of LoRa for vehicular communications. *Comput. Commun.* 2020, 160, 299–310, https://doi.org/10.1016/j.comcom. 2020.06.006.
- [90] Priyanta, I.F.; Golatowski, F.; Schulz, T.; Timmermann, D. Evaluation of LoRa technology for vehicle and asset tracking in smart harbors. In Proceedings of IECON 2019-45th Annual Conference of the IEEE Indu strial Electronics Society, Lisbon, Portugal, 14–17 October 2019, https://doi.org/10.1109/IECON.2019.89 27566.
- [91] Filho, F.H.C.d.S.; Dester, P.S.; Stancanelli, E.M.G.; Cardieri, P.; Nardelli, P.H.J.; Carrillo, D.; Alves, H. P erformance of LoRaWAN for Handling Telemetry and Alarm Messages in Industrial Applications. *Sensor* s 2020, 20, 3061, https://doi.org/10.3390/s20113061.
- [92] Yaye, S.A.R.R.; Gueye, B.; Cheikh, S.A.R.R. Performance analysis of a smart street lighting application u sing LoRa wan. In Proceedings of 2019 International Conference on Advanced Communication Technolog ies and Networking (CommNet), Rabat, Morocco, 12–14 April 2019, https://doi.org/10.1109/COMMNET. 2019.8742356.
- [93] Vitadhani, A.; Alief, F.; Haryanto, B.; Harwahyu, R.; Sari, R.F. Simulating LoRaWAN for flood early war ning system in ciliwung river, bogor-Jakarta. In Proceedings of 2020 International Seminar on Application

for Technology of Information and Communication (iSemantic), Semarang, Indonesia, 19–20 September 2020, https://doi.org/10.1109/iSemantic50169.2020.9234221.

- [94] Mnguni, S.; Mudali, P.; Sibeko, N.; Adigun, M.O. Lora gateway placement at the university of zululand: A case study. In Proceedings of 2019 International Conference on Smart Applications, Communications a nd Networking (SmartNets), Sharm El Sheikh, Egypt, 17–19 December 2019, https://doi.org/10.1109/Sma rtNets48225.2019.9069762.
- [95] Farhad, A.; Kim, D.H.; Pyun, J.Y. Scalability of LoRaWAN in an urban environment: A simulation study. In Proceedings of 2019 Eleventh International Conference on Ubiquitous and Future Networks (ICUFN), Zagreb, Croatia, 2–5 July 2019, https://doi.org/10.1109/ICUFN.2019.8806140.
- [96] Lorawan® regional parameters. Available online: https://lora-alliance.org/wp-content/uploads/2021/05/RP 002-1.0.3-FINAL-1.pdf. (accessed on 8 August 2022)
- [97] Antenna specifications. Available online: https://lora-developers.semtech.com/documentation/tech-papersand-guides/the-book/antenna-specifications/. (accessed on 31 July 2022)
- [98] A technical overview of lora<sup>®</sup> and lorawan<sup>™</sup>. Available online: https://lora-alliance.org/wp-content/uploa ds/2020/11/what-is-lorawan.pdf. (accessed on 8 August 2022)
- [99] The things fundamentals on lorawan. Available online: https://www.thethingsnetwork.org/docs/lorawan/. (accessed on 18 June 2022)